Author's response to reviews

Title: Epidermoid cyst of the coronal sulcus mimicking penile cancer: a case report.

Authors:

Luigi Cormio (luigicormio@libero.it)
Vito Mancini (mancini.uro@gmail.com)
Francesca Sanguedolce (fradolce@hotmail.com)
Paolo Massenio (p.massenio@alice.it)
Livia Fucci (livia.fucci@gmail.com)
Pantaleo Bufo (p.bufo@unifg.it)
Traficante Antonio (traficanteantonio@libero.it)
Giuseppe Carrieri (giuseppe.carrieri@unifg.it)

Version: 2
Date: 5 May 2014

Author's response to reviews:

Dear Ms. Anna Kathrina Vidal

Journal Editorial Office, BioMed Central

Thank you for asking a revised version of our manuscript.

As for your point 1 (please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report), we changed the title as follows:

Epidermoid cyst of the coronal sulcus mimicking penile cancer: a case report.

This should meet your request of A (epidermoid cyst of the coronal sulcus) presenting with (in this case mimicking rather than presenting with) B (penile cancer).

As for your point 2, we included the co-authors' email addresses in the Title page.

As for your point 3, we changed the 'Background' section header to 'Introduction'.

As for your point 4, we included an acknowledgement section at the end of the manuscript before the reference list stating that "this study received no funding".

We hope our manuscript will now be suitable for a fast revision and editorial decision.
Grateful for kind your assistance,
best regards.

Luigi Cormio, MD PhD
Associate Professor of Urology
Director of Endourology Section and Deputy Head of Department
Department of Urology and Renal Transplantation
University of Foggia, Italy